Author's response to reviews

Title: Systematic reviews of complementary therapies - an annotated bibliography. Part 3: Homeopathy

Authors:

Klaus Linde (Klaus.Linde@lrz.tu-muenchen.de)
Maria Hondras (mhondras@interaccess.com)
Andrew Vickers (vickersa@mskcc.org)
Gerben ter Riet (G.terRiet@EPID.UNIMAAS.NL)
Dieter Melchart (Dieter.Melchart@lrz.tu-muenchen.de)

Version: 2 Date: 9 Jul 2001

Dear Biomed Central CAM team,
please find enclosed the revised versions of the homeopathy review of reviews. I will try to resubmit the herbal medicine and acupuncture manuscripts in the next minutes (a first trial to open the acupuncture manuscript failed). A letter with the responses to peer reviewers and summarizing the changes we have made in the 3 manuscripts is to be sent with the acupuncture manuscript.
I hope the revised manuscripts will be acceptable.
Best wishes
Klaus Linde
(This week in Berlin with email Klaus.Linde@charite.de)